Author’s response to reviews

Title: Overexpression of Sterol Carrier Protein 2 in Patients with Hereditary Cholesterol Gallstones

Authors:

YunFeng Cui (yunfengcuidoctor@yahoo.com.cn)
ZhongLian Li (lizhonglian@medmail.com.cn)
ErPeng Zhao (epzhao@yeah.net)
YanFeng Jia (jyf102030@qq.com)
DongHua Li (NKLDH@163.com)
Ju Zhang (zhangju@nankai.edu.cn)
NaiQiang Cui (yfcuink@hotmail.com)

Version: 9 Date: 1 February 2011

Author’s response to reviews:

Dear Editor,

Thank you and the reviewers for the helpful and insightful comments. We have revised the whole manuscript carefully according to reviewer’s comments.

Responses to editorial requests:
1. I add an author, ErPeng Zhao, to this paper because this colleague also provided some samples for us.
2. I have copyedited my paper with the help of a native English speaking colleague.

Responses to Dr. Michael Fuchs’s comments:
The reference 22 has been eliminated.

Responses to Dr. Silvana Zanlungo’s comments:
I have done some English language corrections according to Dr. Silvana Zanlungo’s instructions.

Sincerely yours,

NaiQiang